## **Author's Response To Reviewer Comments**

Clo<u>s</u>e

We have added the figures as separate files, included the GIGADB citation and point the reader to just one github repository now.

We thank the reviewer 1 to point us to the FTP. However, the README in the FTP states that this is a liftover and not a call set for SV. Thus, our point remains.

Clo<u>s</u>e